Wed Apr 9 16:43:33 GMTI 2014

Year: 14 Doy: 099
observer: berkey
WEATHER COMMENT: berkey: Wed Apr 9 16:43:36 GVl 2014
temp 40f, wind 5nph fromthe S, clear skies inversion |layer is capping haze bel ow about 7,000 feet.

end
Wed Apr 9 16:51: 05 GMI 2014: PSPT Start Patrol
GENERAL OBSERVATORY COMMENT BY berkey: Wed Apr 9 16:56:19 GV 2014
One nore |ast day with observing on the old CoMP PC.

end
KCOR COMMENT BY berkey: Wed Apr 9 16:57:45 GV 2014
Connected up the spare fiber Allen pulled down fromthe spar yesterday. Connected to the R caneralink fiber extender and f
i ber ports on the back of the spar. Wth this PFRenmote and Ci View can talk to canera 0. WAiting on steves arival for the
T/ caneral fi ber.
___end
Wed Apr 09 17:02:05 GMI 2014 COW Start Patrol
CoMP COWMENT BY berkey: Wed Apr 9 18:11:40 GV 2014
Testing the copy to m soserver script on new conp conputer. During this test | did multiple copies of a folder naned 2018
5555 with test.txt and test.fts inside. The data is jibberish, but it may show up in LDM and the rest of the pipeline, if
it does plsease either ignore or delete it.

end
Wed Apr 09 18:19:37 GMI 2014 COWP End Patr ol
Wed Apr 09 18:19:38 GMI 2014 COW Start Patrol
Wed Apr 9 19:01:59 GV 2014: PSPT Start Patrol
PSPT COMMENT BY berkey: Wed Apr 9 19:02:03 GV 2014
Moved pspt keyboard and mouse control onto the KVM switch.

end
Wed Apr 09 19:10:51 GV 2014 COWP End Patr ol
GENERAL OBSERVATORY COMMENT BY berkey: Wed Apr 9 19:26:06 GMIr 2014
Looks |i ke neadowl ark controller #2's USB cabl e was not plugged in this norning, so this nmornigns CoMP data may be m ssing
hal f of its nodul ati on.
__end____
KCOR COMMENT BY berkey: Wed Apr 9 23:19:31 GV 2014
Kcor is back online. :)

___end___
CoMP COWMENT BY berkey: Thu Apr 10 02:48:59 GJMI 2014
CoMP meadow ark controllers are still not talking with |labview This nmorning we were able to get wi ndows drivers tonorrow

Hopeful |y tonmorrow we can get new LabView drivers.
end



